Author's response to reviews

Title: Towards an understanding of the information and support needs of surgical adolescent idiopathic scoliosis patients: a qualitative analysis

Authors:

Radha MacCulloch (radha.macculloch@sickkids.ca)
Sandra Donaldson (sandra.donaldson@sickkids.ca)
David Nicholas (david.nicholas@sickkids.ca)
Joyce Nyhof-Young (joyce.nyhof-young@uhn.on.ca)
Ross Hetherington (ross.hetherington@sickkids.ca)
Doina Lupea (doina.lupea@cancercare.on.ca)
James G Wright (james.wright@sickkids.ca)

Version: 4 Date: 1 May 2009

Author's response to reviews: see over
May 1, 2009

To The Scoliosis Editorial Team:

Thank you very much for your final review of the manuscript entitled, Towards an understanding of the information and support needs of surgical adolescent idiopathic scoliosis patients: a qualitative analysis (MS: 6819345512585212).

The authors have completed a final proofing of the manuscript and made a few minor edits. Please see below for a detailed breakdown of the edits made.

Sincerely,

Radha MacCulloch
Sandra Donaldson
David Nicholas
Joyce Nyhof-Young
Ross Hetherington
Doina Lupea
James G Wright